Author’s response to reviews

Title: A systematic review of genetic mutations in pulmonary arterial hypertension

Authors:

Gerardo Garcia-Rivas (gdejesus@itesm.mx)
Carlos Jerjes-Sanchez (carlos.jerjes@udicem.org)
David Rodriguez (hd.rodriguez.f@gmail.com)
Jose Garcia-Pelaez (josegp_92@hotmail.com)
Victor Trevino (vtrevino@itesm.mx)

Version: 1 Date: 30 Dec 2016

Author’s response to reviews:

Following the editor suggestion (below):

(1) We have removed the Table 3 that listed the commercial gene panels related to PAH.

(2) We amended the text related to Table 3 (marked in red in discussion).

(3) We also adapted the Table 2 to include the availability of genes without referencing to commercial gene panels (marked in red).

(4) We replaced "commercial" for "available" when needed (marked in red).

(5) We amended the title of the manuscript.

------------------------

Editor's comments:

This paper lists companies that offer panels for testing for pulmonary arterial hypertension (PAH). This may be perceived as conflict of interest and it would be better to omit this part of the manuscript and resubmit as a review of the genes that can cause PAH.

------------------------